

Oldenburg Discussion Papers in Economics

The effect of intermittent renewable supply on the forward premium in German electricity markets

Marius Paschen

V – 397–16

November 2016

Department of Economics University of Oldenburg, D-26111 Oldenburg

The effect of intermittent renewable supply on the forward premium in German electricity markets

Marius Paschen¹

Abstract

Renewable energy such as wind or solar power currently contributes a large share to the total German electricity supply as a result of the German energy transition. This paper presents an empirical analysis of how power shocks resulting from intermittent renewable supply affect forward premiums in German electricity markets. We contribute to the existing literature by investigating determinants of forward premiums, thereby focusing on wind and on solar power. We find positive wind shock effects on forward premiums. This can be explained as a consequence of the merit order effect. The findings in this paper underline the need to introduce wind power futures at the EEX to reduce the risk mark-up for participants in forward markets. *Keywords:*

Electricity Market, Forward Premiums, Intermittent Wind and Solar Power *JEL Classification:* Q42, Q41, C22, C51

Email address: marius.paschen@uni-oldenburg.de (Marius Paschen)

¹Carl von Ossietzky University Oldenburg, Faculty II, Department of Economics and Laws, Ammerländer Heerstraße 114-118, D-26129 Oldenburg, Germany, Phone: 015735617024

1. Introduction

This paper presents and empirical analysis of the effect of intermittent renewable supply (mainly wind) on forward premiums (see section 2) in German electricity markets. In recent years, there has been a rise in the level of renewable generating capacity in Germany. This is partly the result of the German energy transition (launched by the government's feed-in law of 1991 and formulated in further detail in the Renewable Energy Act (EEG) of 2000). Renewable energy sources such as wind power currently contribute a large share to total German electricity supply. In 2015, renewables accounted for 30.1% of the total German electricity supply. The most important renewable energy source was wind power, with a 44.9% share of the total supply of renewables. Solar (19.6%) and hydro (10%) made up smaller but significant shares (BDEW (2016)). A priority feed-in for renewables exists.

Spot prices depend negatively on wind and solar power. This refers to the merit order effect (cf. Jensen and Skytte (2002), Fischer (2010), Würzburg et al. (2013) and Cludius et al. (2014)).

We therefore argue that important renewable power variables should be taken into account when analyzing forward premiums in German electricity markets. We contribute to the existing literature by empirically investigating determinants of the forward premium in German electricity markets. We focus on intermittent wind and solar power because there is very little research on the effects of these potentially important drivers on forward premiums. In particular, to our knowledge there are no empirical studies that investigate German electricity markets, despite the high share of renewable feed-in. As in Botterud et al. (2010) our aim is not to present a new theory for forward premiums: instead, we want to empirically analyze the extent to which wind and solar power can explain the forward premium.

Electricity wholesale markets are characterized by a need to match demand and supply precisely at every point in time, despite supply and demand shocks. In addition, short-term demand is very price-inelastic, and power cannot be stored. Financial risk management by such means as long-term contracts is, therefore, an important tool for reducing price risks when facing such volatile markets. The European Energy Exchange (EEX), one of the largest European power exchanges, uses such contracts in forward markets to determine electricity forward prices for specific time periods in the future to hedge against risks of future day-ahead spot prices. The difference between forward and spot prices at the time to delivery is the forward premium.

In this paper, we report significant positive wind shock effects on both monthly and daily ex-post forward premiums. The wind shock shifts the supply curve to the right due to the merit order effect at the time when the spot price is realized. The spot price should fall as a result. Furthermore, it should fall below the current forward price because the unexpected future wind shock cannot be accurately taken into account in the forward price. Therefore, forward premiums should rise. The findings we report here underscore the need for the EEX strategy to introduce a new wind power future. This is a new market instrument designed to provide market participants with the opportunity to hedge specifically against price and quantity risks of wind power generation. It would then reduce the risk mark-up for participants in forward markets resulting from intermittent wind feed-in as further discussed in sections 4.1 and 6. The wind shock effect on daily forward premiums is substantially lower, i.e., near zero. Substantially reducing the maturity of the forward contract to the very near future is likely to result in almost perfect approximations of the expected spot price at the time when the forward contract is determined. This should lead to forward premiums as well as wind shock effects on the forward premium near zero. We find no significant solar shock effect on daily premiums. This again might be due to forward premiums close to zero. The wind shock effect on monthly peak load premiums is larger than on base load premiums. Intuitively, at peak load times, there is a higher demand for power. Thus, more power supply (normally with gas as the back-up generation type) is needed, leading to a higher equilibrium spot price. If there is then an unexpected wind shock at peak load times, the spot price might fall further below the current forward price due to higher differences in marginal costs at the right of the merit order curve, resulting in larger forward premiums.

This paper is structured as follows. Section 2 presents past research on forward premiums. Section 3 presents the chosen model for estimation as well as related data and variables. In section 4, estimation results are presented. Section 5 shows some robustness checks. Section 6 concludes.

2. Literature review

Electricity is not storable and is a flow rather than a stock. Thus, classic cost-of-carry approaches for spot and forward markets, which date back to Kaldor (1939), are not applicable (cf. Bessembinder and Lemmon (2002)). Equilibrium approaches are used instead to model the relationship between spot and forward prices (dating back to Keynes (1930)) when market agents build expectations about future spot prices (cf. Redl et al. (2009)). Agents with the need to hedge spot price uncertainty treat the forward price as being determined as the expected future spot price plus an ex-ante premium. This premium is, unfortunately, unobservable. Thus, empirical research often looks instead at the ex-post forward premium, which is the difference between the forward price and the ex-post spot price at the time to delivery. The relation between the two premiums is the following:

$$forward_{t,t+k} - spot_{t+k} = forward_{t,t+k} - E_t[spot_{t+k}] + E_t[spot_{t+k}] - spot_{t+k}$$
$$= forward_{t,t+k} - E_t[spot_{t+k}] + \epsilon_t.$$
(1)

The forward price, determined at time t contracting delivery for time t + k(k is the contract maturity), minus the realized spot price at time t + k is the ex-post premium. It is equal to the difference between the forward price and the expectation of the spot price at time t + k (called "future spot price" in the remainder of this paper) made at time t plus the difference between the expected future spot price and the realized spot price. The first difference is the ex-ante premium. The second is the error term ϵ_t . It refers to price shocks between t and t + k (which may occur due, for example, to price-inelastic short-term demand). The ex-post premium is a consistent estimator of the ex-ante premium if the error distribution has zero mean (cf. Redl and Bunn (2013)).

One way to model equilibria in forward markets is to create a two-stage game where Cournot producers behave more competitively in the spot market due to forward commitments they have made in an oligopolistic environment (cf. Allaz and Vila (1993)). Further model assumptions are risk neutrality of agents and no arbitrage (leading to a zero ex-ante premium). In Ritz (2016), the model of Allaz and Vila (1993) was expanded to incorporate intermittent renewable generation and provide solutions also for more than two strategic players. However, results of this game theory approach differ when the twostage assumption is relaxed. Therefore, the results are ambiguous (cf. Redl and Bunn (2013)). Furthermore, Bessembinder and Lemmon (2002) stated that the no-arbitrage condition might not hold when the good under investigation is non-storable electricity.

This leads to the other approach to equilibrium modeling of forward markets. Bessembinder and Lemmon (2002) created an equilibrium model with riskaverse utility maximizing agents by applying a competitive marginal-costbased approach (producers equate their marginal revenues to their marginal costs) for specific electricity forward and spot markets. However, a weakness of this model is that demand is an exogenous random variable and not derived by the model itself. One solution that has been proposed in the literature is to endogenously determine the ex-ante premium as a linear function of the expected variance (negative effect) and expected skewness (positive effect) of spot prices, and a large body of empirical literature has sought to verify this result. The stated moments of the spot price distribution can be seen as measures of risk management of the market participants. When participants believe that future spot prices will be unusually high (i.e., expect a high skewness), they will determine contracts with higher forward prices, such that the forward premium goes up (cf. Weron and Zator (2014)). On the other hand, when market participants believe that future spot prices will vary widely (i.e., expect higher risk and a high variance), forward prices and, therefore, forward premiums will decrease. Peura and Bunn (2016) developed a model of spot and forward markets combining the hedging aspects of Bessembinder and Lemmon (2002) and the strategic aspects of Allaz and Vila (1993). They also analyze the impact of intermittent wind on these markets. Gersema and Wozabal (2016) extended the model of Bessembinder and Lemmon (2002) by including wind power futures. Using a simulation, they found that market actors benefit from these futures due to specific hedging against wind power risks.

Many empirical analyzes have found different results when attempting to verify the derived functional relationship between the forward premium and the stated spot price distribution moments of Bessembinder and Lemmon (2002) (BL-relation). Empirical results vary across markets and chosen timescales for the premium. Furthermore, depending on the market, other drivers have been identified as important. For example, Botterud et al. (2010) analyzed weekly risk premiums (negation of forward premiums) in the most mature hydro-dominated electricity market in the world (Nord Pool). They could not confirm the BL-relation, but found that variables referring to hydro power are further important drivers of the risk premium. At Nord Pool, hydro power (with marginal costs equal or close to zero) contributes the largest share to total electricity supply, which stands in contrast to the marginal-cost-based approach of Bessembinder and Lemmon (2002). Weron and Zator (2014) found a different effect of water reservoir levels on the risk premium as in Botterud et al. (2010) when revisiting the empirical research of forward and spot markets at Nord Pool due to potential empirical pitfalls (see section 5). They still could not confirm the BL-relation. Woo et al. (2015) found a positive effect of wind power on hourly forward premiums when analyzing Californian day-ahead and real-time electricity markets of the United States (US). In Woo et al. (2016), merit order effects are found and further analyzed in these markets. Redl et al. (2009) analyzed monthly forward premiums in German markets (among others) managed by the EEX. They could not confirm the negative effect of the variance of spot prices, but the positive effect of the skewness of spot prices on the forward premium. Moreover, in addition to drivers regarding risk assessment, they included drivers that refer to supply and demand in order to control for related shocks influencing spot prices. Redl and Bunn (2013) expanded this research to analyze these premiums using a multi-factor model, and found other important drivers in these markets, such as forward premiums of gas production and a variable combining supply and demand shocks called the margin shock. They found a positive effect of skewness and of a volatility factor of spot prices on the forward premium (the latter stands in contrast to the result reported by Bessembinder and Lemmon (2002)). Fleten et al. (2015) developed and empirically tested a model for forward premiums by analyzing forward contracts directly using Nord Pool and EEX data (without taking into account renewable drivers). The results supported the BL-relation. They stated that ex-post premiums including forward contracts with longer maturities than one day are hard to interpret because financial traders in forward markets want to make profit from their positions. Therefore, they do not necessarily hold the contracts throughout the whole maturity. Thus, ex-post premiums measure compensation for risk, assuming that contracts will be held until their delivery date, which in reality is not necessarily the case. Forward premiums in EEX markets were also analyzed by, for example, Diko et al. (2006), Kolos and Ronn (2008) and Benth et al. (2008).

3. Estimating forward premiums

3.1. Overview

German electricity markets have become even more volatile as the feedin of high fluctuating renewables has increased. We find here that forward premiums significantly differ from zero, which indicates that the no-arbitrage condition (cf. section 2) does not hold in current German electricity forward markets. Therefore, the BL-relation of Bessembinder and Lemmon (2002) explaining the forward premium is used as a starting point for estimation, with moments of the spot price distribution as explanatory variables to control for risk assessment. However, these drivers should only be able to explain premiums in recent German electricity markets to a limited extent, because marginal costs of renewables such as wind or solar power are also near or equal to zero and therefore stand in contrast to a marginal-cost-based approach. Thus, drivers of supply and demand (with a focus on wind power) are included to control for their related shocks. The variables are chosen according to the multi-factor analysis of premiums in Redl and Bunn (2013) because they empirically investigate determinants of ex-post forward premiums in German electricity forward and spot markets in a detailed way (without focusing on renewable power). The BL-relation explaining the forward premium is, as stated in section 2, a linear function. Therefore, effects on the ex-post forward premium are estimated using ordinary least squares (OLS). If the OLS residuals have zero mean, the ex-ante premium should be estimated consistently, and separation between the premium drivers and the error term is possible (cf. Redl and Bunn (2013)).

We use time series of monthly as well as of daily forward contracts of EEX German electricity markets to analyze month-ahead as well as day-ahead premiums. There exist many monthly as well as daily contracts, and both are therefore relevant to investigate. Furthermore, we want to investigate whether there is a significant difference in the wind shock effects on the forward premium when the maturity time shrinks from a monthly to a daily timescale. Another argument for analyzing ex-post day-ahead premiums is that interpretation is possible because traders cannot short positions throughout maturities of only one day. For analysis of ex-post month-ahead premiums in this paper, interpretation is only possible if the assumption that traders hold positions through the whole maturity (cf. Fleten et al. (2015), see section 2) is not violated, as in past empirical research.

We use time series of the last few years: This is a period in which renewables came to play an important role in German electricity markets as a result of the energy transition. Solar shock effects can only be investigated for daily premiums due to data availability. We further consider premiums including both base and peak load prices in order to analyze whether there are different renewable shock effects on the forward premium at peak load times.

3.2. Monthly data

The time period from May 2009 to December 2015 includes 80 monthly observations. We use Phelix one-month forward prices of the contracts made on the last trading day in every month for the following month. When using monthly data, forecast errors referring to month-ahead premiums should be as low as possible (cf. Redl et al. (2009)). The last trading day is chosen in order to use the best expectation of the spot price of the next month of market participants as in Redl and Bunn (2013). Spot prices are average monthly prices of hourly time series of day-ahead spot prices traded at the EEX.² The variable to be explained, both for base and peak load,³ is the relative ex-post forward premium, which is the ex-post forward premium defined in section 2 divided by the realized future spot price:

$$premium_{t,t+1} = \frac{forward_{t,t+1} - spot_{t+1}}{spot_{t+1}},$$

$$premium_peak_{t,t+1} = \frac{forward_peak_{t,t+1} - spot_peak_{t+1}}{spot_peak_{t+1}}.$$
(2)

Figure 1 shows time series of forward prices, realized one month future spot prices (*spot_ahead*), and forward premiums. Starting in May 2009, both base and peak load spot prices increase over time up to the end of 2010. From the beginning of 2011 on, they decrease, which should be due to the merit order effect resulting from renewables. The time series also clearly indicate seasonal behavior, with higher prices in months with higher demand (autumn and winter). Peak load prices are higher than base load prices due to higher demand at peak hours of the day. The behavior of spot prices over time is relatively well anticipated by the forward prices. For peak load, for-

²Data source for Physical Electricity Index (Phelix) future and spot prices for the market area of Germany and Austria is EEX (http://www.eex.com/de/). For the investigated time period, many monthly forward contracts with different monthly maturities are set every day. We use forward settlement prices (\in /MWh) of the last trading day for the next month in every month. Original spot prices are day-ahead settlement prices (\in /MWh) for each hour of the next day.

³For base load, the underlying for daily forward prices are daily averages of hourly day-ahead spot prices (Phelix Day Base). For peak load, the underlying are daily averages of hourly day-ahead peak spot prices (Phelix Day Peak) considering only peak hours from 8:00 am to 8:00 pm of the next day.



Y-axis: Forward, spot prices (\in /MWh) and relative premiums (base (upper graphics) and peak load (lower graphics)).

Figure 1: Monthly time series forward, one month future spot prices and forward premiums

ward prices overestimate spot prices more than for base load, and there is no underestimation. Therefore, the peak load forward premium is higher than for base load and is always positive. For base load, there are also months of underestimating the spot price, resulting in negative premiums. In contrast to Redl and Bunn (2013), time fixed effects modeled as quarterly and yearly time dummies are included in the regressions to capture seasonal behavior in the price series as well as in several following explanatory variables.⁴

Table A.7 in Appendix A presents descriptive statistics of both base and peak forward premiums and of potential explanatory variables. Both premium means are positive, mean and standard deviation of the peak premium are

 $^{^4\}mathrm{Seasonality}$ in the price series may be transferred to the premiums, the variables to be explained.

higher than those of the base premium as in Redl and Bunn (2013). This clearly indicates the existence of non-zero premiums and, therefore, the invalidity of the no-arbitrage condition in German markets, as discussed in section 3.1. Furthermore, non-zero premium means are significant (cf. again table A.7 in Appendix A). For base premiums, forward prices are on average traded lower above spot prices on the last monthly trading day, but for peak premiums substantially higher. For both premiums there is less volatility than in Redl and Bunn (2013).

One explanatory variable of interest is the wind margin, which is wind power divided by total load. Such a combined variable is used to control for both supply and demand (cf. Redl and Bunn (2013)). The difference is that supply looks only at the specific renewable type of wind generation to analyze the effect of wind supply alone on forward premiums. Current values of this variable in the same month when the forward contract is set are observable by market participants. However, wind and load shocks occurring in t + 1are not, which is captured in the one future month margin variable, the other variable of interest. In addition,⁵ hydro margin (hydro power divided by load) and its shock are further variables referring to a specific renewable

⁵Data source for wind, hydro power, and load is the European Network of Transmission System Operators for Electricity (ENTSO-E) (http://www.entsoe.eu/). Load is average German total demand for power (GW) per month, including domestic load, net exports (exports minus imports, all electricity trades between respective countries), and network losses. Wind and hydro are average German wind power and, respectively, hydro power feed-in (GW) per month. In hydro, all types of hydro power available for German power generation are included.

supply type using this combined approach:

$$margin_wind_{t} = \frac{wind_{t}}{load_{t}}, \ margin_wind_shock_{t} = \frac{wind_{t+1}}{load_{t+1}},$$
$$margin_hydro_{t} = \frac{hydro_{t}}{load_{t}}, \ margin_hydro_shock_{t} = \frac{hydro_{t+1}}{load_{t+1}}.$$
(3)

According to Redl and Bunn (2013) these variables stand for fundamental and shock effects on forward premiums (focusing on specific supply types). Due to robustness checks, wind, hydro and load series are taken into account separately (see section 5). Other important variables in this context should be margin variables referring to intermittent solar power. However, not enough observations are available on a monthly timescale (data available starting in January 2011). Considering table A.7 in Appendix A, wind power covers on average about 10% of total load, hydro about 4%. Volatility for wind is relatively high due to its inherent intermittency in contrast to hydro power, which in principle is constant. Figure 2 presents time series of wind, load, and the wind margin. In both wind and load series, the same seasonal behavior is observable as in the price series. Load and wind is higher in autumn and winter months. In the load series, there might also be yearly cyclical behavior. Wind power is rising over time due a higher focus on renewables in Germany and the priority feed-in. The behavior of wind is in principle transferred to the wind margin.

Further explanatory control variables are moments of the spot price distribu-



Y-axis: Wind power (GW), load (GW) (upper graphics), wind margin (lower graphic).

Figure 2: Monthly time series wind power, load and wind margin

tion (also of higher order): Mean, standard deviation, skewness and kurtosis of hourly spot prices in every current month (notation: *spot* for the mean and *spot_moment* for the others). Standard deviation as a volatility factor and skewness are already included in the model of Bessembinder and Lemmon (2002) as drivers on the forward premium, as stated in section 2. Except for the mean, these variables refer to behavioral effects on forward premiums as reported in Redl and Bunn (2013). The authors argue that the spot price kurtosis on the premium should also have a positive effect due to increasing interest in fat tails and aversion to extreme outcomes of spot prices when assuming adaptive price expectations of market participants. In addition, the authors argue that the volatility effect could also be positive because, due to the merit order convexity, shocks creating high skewness and volatility could have the same signs. Table A.7 in Appendix A shows descriptive statistics of spot price controls. Monthly equilibrium spot price is about 40 \in /MWh on average with a high standard deviation. Spot standard deviation and kurtosis are relatively high on average; kurtosis also has a high standard deviation. Spot skewness is very low, but negative on average.

A further control variable is the *basis*, the difference between current forward (last trading day) and spot (monthly average) prices, referring to dynamic effects on forward premiums as in Redl and Bunn (2013). When forming expectations about month-ahead spot prices, market agents may adapt these expectations to recent spot price averages. The effect of the basis on the forward premium should therefore be positive. Forward prices are on average traded above current spot prices on the last trading day of the current month, as shown in table A.7 in Appendix A, but there is also moderate standard deviation.

Table A.11 in Appendix A shows cross-correlations between explanatory variables.

All variables are tested for stationarity using the augmented unit root test of Dickey and Fuller $(1979)^6$ (cf. table A.9 in Appendix A). All variables are chosen to be stationary and, therefore, their levels are used.

⁶The process under the null is assumed to be an autoregressive process of order 1 (AR(1)). By graphical inspection, it is not clear which test equation to choose. Therefore, different test equations are used and test decisions with a majority are selected. The time series of wind and wind margin indicate trends. However, test results indicate stationarity in levels for wind power and wind margin, especially when a trend is included. Thus, these variables are selected to be stationary.

3.3. Daily data

The time period is November 22, 2012 to December 30, 2015, including 778 daily observations. We use Phelix one-day forward prices made one trading day before the following day. At EEX, day-ahead forward contracts have the shortest maturity. Besides following the argument that this keeps corresponding forecast errors as low as possible, we want to compare monthahead premiums with premiums when the maturity of the forward contract is as short as possible. Furthermore, according to Fleten et al. (2015), dayahead premiums are not problematic to interpret as stated in section 2. At the EEX, forward contracts with maturities of one day are a relatively new financial tool. These contracts were introduced at the end of 2012 (data available starting on November 22, 2012). Therefore, the time period as a whole is about 3 years shorter for these data than for the data on monthahead premiums. However, renewable feed-in has been present in Germany especially in the last few years and, thus, this time period is also relevant. Dependent variable is the relative ex-post forward premium as in section 3.2, now using daily data:

$$premium_{t,t+1} = \frac{forward_{t,t+1} - spot_{t+1}}{spot_{t+1}},$$

$$premium_peak_{t,t+1} = \frac{forward_peak_{t,t+1} - spot_peak_{t+1}}{spot_peak_{t+1}}.$$
(4)

Spot prices are the daily indices (Phelix Day Base, Phelix Day Peak), the average base and peak daily prices of hourly time series of day-ahead spot



Y-axis: Forward, spot prices (\in /MWh) and relative premiums (base (upper graphics) and peak load (lower graphics)).

Figure 3: Daily time series forward, one day future spot prices and forward premiums

prices traded at the EEX.⁷

Figure 3 shows time series of forward prices, future spot prices (*spot_ahead*), and forward premiums. Decreasing prices due to the merit order effect in that time period are also present, as well as seasonal behavior (cf. section 3.2). In addition, weekly behavior might also be observable. By graphical inspection of the price series, no difference in the forward and realized future spot prices is observable, resulting in forward premiums very close to zero. It seems that approximations of one-day future spot prices are nearly perfect. However, as shown in table A.8 in Appendix A, Wald test results

⁷Data source for Phelix future and spot prices is again EEX, as in section 3.2. We use forward settlement prices (\in /MWh) for the next day at (only) every trading day. Original spot prices are day-ahead settlement prices (\in /MWh) for each hour of the next day.

indicate that the premiums are significantly different from zero. Therefore, analyses of day-ahead premiums are considered in this paper as well, despite premiums near zero. Time fixed effects modeled as weekly, monthly, and yearly time dummies are included in the OLS regressions in order to capture seasonal behavior in daily price series as well as in several subsequent daily explanatory variables. Table A.8 in Appendix A shows descriptive statistics of the premiums and of explanatory variables. Descriptive moments and statistics of the premiums are zero.

Main explanatory variables⁸ are again the margin variables and their shocks (cf. section 3.2) for daily data. Production types are now wind and solar:

$$margin_wind_{t} = \frac{wind_{t}}{load_{t}}, \ margin_wind_shock_{t} = \frac{wind_{t+1}}{load_{t+1}},$$
$$margin_solar_{t} = \frac{solar_{t}}{load_{t}}, \ margin_solar_shock_{t} = \frac{solar_{t+1}}{load_{t+1}}.$$
(5)

German solar data is available starting in January 2011 as stated in section 3.2. Therefore, enough observations to conduct regressions are available when using daily data. Separate wind, solar, and load series are analyzed in section 5. Hydro data are not available on a daily timescale. Considering table A.8

⁸For wind and solar data sources are the German transmission system operators (http://www.amprion.net,http://www.tennettso.de,http://www.50hertz.com/ de,http://www.transnetbw.de). Original data is average German power feed-in (MW) per 15 minutes. Few days are omitted due to data gaps on specific days from specific sources. Data is transformed into average power feed-in (GW) per day. For load data, the source is again ENTSO-E (http://www.entsoe.eu/). Original load is average German total demand (cf. section 3.2) for power (MW) per hour. Data is transformed into average demand (GW) per day.

in Appendix A, wind power covers on average about 10% of total load also for this time period on a daily timescale. Solar covers about 6%. Volatility for wind is now higher compared to monthly data due to also intermittent feed-in on different days. Solar power also deviates from the mean to a relatively large extent due to its intermittency, but only amounts to half of wind volatility. Figure 4 shows daily time series of wind, wind margin, solar, solar margin, and load. Again, wind and load is higher in the autumn and winter months. Solar is, in contrast, higher in spring and summer due to more days of solar power available in those months. Rising wind power over time is again observable. Solar power is relatively constant. In daily load, there might also be a cyclical weekly behavior in addition to possible yearly cyclical behavior present. The behavior of wind and solar are transferred to the margin variables.

Control variables referring to the spot price distribution are the same as for monthly data: Mean, standard deviation, skewness, and kurtosis of hourly spot prices, but now on every current day. Considering descriptive statistics as shown in table A.8 in Appendix A, for this time period, the daily equilibrium spot price is about $38 \in /MWh$, also with a high standard deviation compared to the monthly spot price. Average spot standard deviation is still high, in contrast to average kurtosis, compared to their monthly averages. Their daily average standard deviations are much lower, especially the one for kurtosis. Daily average spot skewness is very low, but now positive. Finally, the basis control variable, the difference between current day-ahead



Y-axis: Wind power (GW), solar power (GW), load (GW) (upper graphics), wind margin, solar margin (lower graphics).

Figure 4: Daily time series wind power, solar power, load, wind margin and solar margin

forward and spot (daily average) prices, is included. Day-ahead forward prices are on average traded below current spot prices on the current trading day in contrast to prices on a monthly timescale as presented above, but the standard deviation on a daily timescale is twice as high, as shown in table A.8 in Appendix A.

Table A.12 in Appendix A shows cross-correlations between explanatory variables.

All daily variables are also tested for stationarity using the augmented unit root test of Dickey and Fuller (1979) (see section 3.2, cf. table A.10 in Appendix A). All variables are chosen to be stationary and, therefore, their levels are used.

4. Results

4.1. Month-ahead premium

The following general equation is estimated:

$$premium_{t,t+1} = \beta_0 + \beta_1 spot_t + \beta_2 spot_std.dev_t + \beta_3 spot_skewness_t + \beta_4 spot_kurtosis_t + \beta_5 margin_wind_t + \beta_6 margin_wind_shock_t + \beta_7 margin_hydro_t + \beta_8 margin_hydro_shock_t + \beta_9 basis_t + \sum_{i=2}^{4} \gamma_i quarter_{i,t} + \sum_{j=2}^{7} \delta_j year_{j,t} + \epsilon_t.$$
(6)

The sum of $quarter_i$ and $year_j$ are quarterly and yearly time dummies. Table 1 shows regression results for both base and peak load month-ahead premiums. For regressing the base load premium the information criterion of Akaike (1974) (AIC) is minimized when using the mean and the kurtosis of the spot price, wind and hydro margin shocks and the basis as explanatory variables. The very high correlation between spot price skewness and kurtosis should not lead to multicollinearity. The skewness variable is insignificant in both base and peak regressions with and without including spot kurtosis and, therefore, the insignificance is not due to omitted variable bias. Moreover, multicollinearity should be not the case because spot skewness is also insignificant when spot kurtosis is omitted.⁹ Thus, regressions without spot

 $^{^{9}}$ Following Verbeek (2008) multicollinearity might be a problem if the modulus of a cross-correlation between explanatory variables (cf. Table A.11 in Appendix A) exceeds 0.8.

VARIABLES	premium	premium	premium	premium_peak	premium_peak	premium_pea
spot	0.00699***	0.00786***	0.00586**	0.00596**	0.00697**	0.00439
	(0.00211)	(0.00242)	(0.00276)	(0.00294)	(0.00309)	(0.00318)
spot_std.dev.	0.00194	0.00277	0.00272	0.00466	0.00581*	0.00575*
	(0.00242)	(0.00326)	(0.00269)	(0.00291)	(0.00313)	(0.00302)
spot_kurtosis	0.00120***		0.00206***	0.00146**		0.00266**
	(0.000228)		(0.000648)	(0.000630)		(0.00114)
margin_wind	-0.481	-0.545	-0.504	-0.413	-0.499	-0.446
-	(0.338)	(0.333)	(0.350)	(0.399)	(0.412)	(0.398)
margin_wind_shock	1.416***	1.454***	1.361***	1.702***	1.746***	1.626***
0	(0.197)	(0.233)	(0.180)	(0.339)	(0.351)	(0.343)
margin_hydro	-1.663	-2.048	-1.720	-0.163	-0.665	-0.242
	(1.380)	(1.347)	(1.588)	(2.294)	(2.357)	(2.284)
margin_hydro_shock	6.809***	7.042***	6.739***	4.063*	4.357**	3.966*
	(1.180)	(1.163)	(1.246)	(2.118)	(2.177)	(2.109)
basis	0.00717***	0.00708***	0.00813***	0.00575^{*}	0.00573	0.00709**
	(0.00243)	(0.00214)	(0.00207)	(0.00330)	(0.00353)	(0.00346)
spot_skewness		-0.0101	0.0149		-0.0114	0.0208
		(0.00853)	(0.0121)		(0.00939)	(0.0166)
Constant	-0.591^{***}	-0.609***	-0.554***	-0.328*	-0.347^{*}	-0.276
	(0.118)	(0.108)	(0.138)	(0.170)	(0.178)	(0.175)
Quarterly dummies	Yes	Yes	Yes	Yes	Yes	Yes
Yearly dummies	Yes	Yes	Yes	Yes	Yes	Yes
F(p-value)	0.000	0.000	0.000	0.000	0.000	0.000
R-squared	0.000			0.621	0.598	0.631
01	-		=0			

skewness are considered instead for interpretation.

Standard errors of Newey and West (1987) in parentheses in base load regressions in order to control for autocorrelated residua

Non-robust standard errors in parentheses in peak load regressions *** p<0.01, ** p<0.05, * p<0.1.

Table 1: Drivers of the month-ahead base and peak load premium

The positive effect of the wind margin shock on the base load forward premium can be interpreted in line with the argument in Redl and Bunn (2013) regarding misjudgments of future supply. The effect refers to a wind shock and not to a load shock as will be shown in the robustness checks. The wind shock (a specific supply-type shock) included in the wind margin shock shifts the supply curve to the right due to the merit order effect at the time when the spot price is realized. Because of this, the spot price should fall. Furthermore, the spot price should fall below the forward price determined one month before because the unexpected wind shock one month ahead cannot be taken into account in the forward price accurately. However, as forward prices are designed to anticipate expected spot prices, they may fall as well. The rise in the premium can also be interpreted as a risk mark-up on forward prices resulting from higher intermittent wind feed-in and related increased spot price volatility.

The positive sign of this effect is in line with the one in Woo et al. (2015) as stated in section 2. It is also in line with Woo et al. (2016), who identified that forward premiums depend on wind forecast errors. Parts of the wind shock variable in this paper should be attributed to these forecast errors. Considering the theoretical model of Ritz (2016), the forward price falls in any case if there is an increase in intermittent renewable power. This increase leads to a displacement of power production in the spot market (merit order effect) and to lower incentives to make forward commitments (forward-contracting effect). The spot price falls only when renewable penetration is sufficiently high, i.e., when the merit order effect outweights the forward-contracting effect. Therefore, this large penetration should be (on average) the case in the investigated German electricity markets. However, some model assumptions in Ritz (2016) such as the no-arbitrage condition are likely to be violated as stated in sections 2 and 3.1. In the model of Peura and Bunn (2016), where a no-arbitrage assumption is not included, forward premiums are also likely to increase due to a wind increase when expected wind power is high and the renewable subsidy scheme is a feed-in tariff.

Spot price volatility, one theoretical effect on the forward premium according to Bessembinder and Lemmon (2002), is not significant. The effect of the spot price kurtosis as a higher distribution moment and the basis effect is positive and significant and, therefore, in line with arguments of Redl and Bunn (2013). These effects have been discussed in section 3.2. In addition, there is a positive significant effect of the current spot price mean on the forward premium. Intuitively, forward prices should take into account the current spot price when trying to anticipate expected future spot prices. Therefore, the forward premium should rise when the current spot price rises.

Regarding the month-ahead peak load premium, the AIC is minimized when using the same explanatory variables and in addition the standard deviation of the spot price. However, the standard deviation is an insignificant driver when considering the regression without spot skewness.

Again, there are positive wind margin shock effects on the peak load forward premium. The wind margin shock effect on the peak load premium is slightly larger than on the base load premium. At peak load times, there is a higher demand for power. Thus, more power supply (normally with gas as the back-up generation type) is needed, leading to a higher equilibrium spot price. If there is an unexpected wind shock at peak load times, the spot price might fall further below the forward price due to higher differences in marginal generation costs at the right of the merit order curve. Therefore, there should be a larger rise in the forward premium. The other control effects are more or less the same as in the base load regression.

When comparing the empirical results reported by Redl and Bunn (2013) for EEX markets, the peak regression fit considering the R-squared of around 0.6 in this paper is relatively high (0.25 in Redl and Bunn (2013)). In this paper, there is also a significant positive basis effect. The skewness effect is insignificant. This stands in contrast to the effect for peak load reported by Redl and Bunn (2013), who found no significant spot price kurtosis effects but a significant spot price volatility effect for base load premiums. These differences may be due to the much lower renewable power feed-in during the earlier time period from November 2003 to December 2009. In this paper, forward premiums are probably not explained by margin effects of the current month, in contrast to the results reported in Redl and Bunn (2013) (see also section 5). However, the latter defined the margin as the share of total supply in the total load. Besides the argument of the different time period, there should be no problem taking current wind power production into account in the forward contract determined in the same month and, therefore, no effect on the forward premium.

Table A.13 in Appendix A shows that the residuals of the month-ahead regressions have zero mean and, therefore, ex-ante premiums should be estimated consistently, as stated in section 2.

4.2. Day-ahead premium

For both base- and peak load premiums the AIC is minimized when using only spot price standard deviation, wind shock, and solar margin as explanatory variables in the regression. Table 2 shows the regression results when all drivers are included as well as when only choosing the variables suggested by minimized AIC. The very high correlation between solar margin and its shock (cf. table A.12 in Appendix A) should not lead to multicollinearity because the solar margin shock is insignificant in all regressions, whether with or without solar margin (see also section 4.1). Thus, only regressions omitting the solar margin shock are considered for interpretation. The following general equation is estimated:

 $premium_{t,t+1} = \beta_0 + \beta_1 spot_t + \beta_2 spot_std.dev._t + \beta_3 spot_skewness_t$

 $+ \beta_4 spot_kurtosis_t + \beta_5 margin_wind_t + \beta_6 margin_wind_shock_t$

 $+ \beta_7 margin_solar_t + \beta_8 margin_solar_shock_t + \beta_9 basis_t$

$$+\sum_{i=2}^{7} \gamma_i week_{i,t} + \sum_{j=2}^{12} \delta_j month_{j,t} + \sum_{k=2}^{4} \eta_k year_{k,t} + \epsilon_t.$$
(7)

The sum of $week_i$, $month_j$ and $year_k$ are weekly, monthly and yearly time dummies.

VARIABLES	premium	premium	premium	premium	premium	premium	premium_peak	premium_peak	premium_peak	premium_peak	premium_peak	premium_peak
spot	-1.14e-06	-7.24e-07	-9.73e-07				2.62e-07	1.46e-07	3.34e-07			
1	(1.07e-06)	(1.06e-06)	(1.08e-06)				(9.08e-07)	(8.80e-07)	(8.98e-07)			
spot_std.dev.	2.68e-06*	2.61e-06*	2.61e-06*	1.73e-06	1.98e-06*	1.70e-06	1.61e-06	1.57e-06	1.58e-06	1.89e-06	1.72e-06	1.87e-06
1	(1.44e-06)	(1.45e-06)	(1.46e-06)	(1.14e-06)	(1.12e-06)	(1.14e-06)	(1.46e-06)	(1.47e-06)	(1.46e-06)	(1.31e-06)	(1.30e-06)	(1.31e-06)
spot_skewness	1.41e-06	-2.09e-06	1.12e-06	. ,		. ,	-7.67e-06	-5.37e-06	-7.80e-06	. ,	. ,	. ,
	(7.87e-06)	(7.63e-06)	(7.90e-06)				(6.86e-06)	(6.81e-06)	(6.92e-06)			
spot_kurtosis	-4.64e-06	-4.27e-06	-4.62e-06				1.36e-06	1.11e-06	1.37e-06			
	(3.33e-06)	(3.33e-06)	(3.39e-06)				(4.02e-06)	(3.97e-06)	(4.02e-06)			
margin_wind	-1.68e-05	2.07e-05	-2.71e-05				3.44e-05	-6.31e-06	2.98e-05			
	(8.87e-05)	(8.94e-05)	(8.95e-05)				(8.41e-05)	(8.30e-05)	(8.43e-05)			
margin_wind_shock	6.00e-05	5.28e-05	8.29e-05	9.79e-05**	0.000101^{**}	0.000102^{**}	8.40e-05	0.000117	9.42e-05	0.000104^{**}	0.000106^{**}	0.000105^{**}
	(7.58e-05)	(7.98e-05)	(7.82e-05)	(4.85e-05)	(4.86e-05)	(4.86e-05)	(8.00e-05)	(7.99e-05)	(7.92e-05)	(4.18e-05)	(4.21e-05)	(4.21e-05)
margin_solar	-0.000242		-0.000330*	-0.000212		-0.000313*	0.000288*		0.000249	0.000209		0.000172
	(0.000166)		(0.000192)	(0.000148)		(0.000180)	(0.000166)		(0.000203)	(0.000151)		(0.000200)
basis	-3.10e-07	-3.60e-07	-1.30e-08				-1.86e-07	2.07e-07	-5.52e-08			
	(9.88e-07)	(1.04e-06)	(1.02e-06)				(1.01e-06)	(9.62e-07)	(9.41e-07)			
margin_solar_shock		1.33e-05	0.000169		2.53e-05	0.000189		0.000192	7.46e-05		0.000158	6.82e-05
		(0.000137)	(0.000162)		(0.000128)	(0.000160)		(0.000161)	(0.000199)		(0.000155)	(0.000206)
Constant	2.21e-05	-4.29e-06	1.36e-05	-3.31e-05	-3.82e-05	-3.26e-05	-6.60e-05	-5.62e-05	-6.98e-05	-4.77e-05*	-4.45e-05	-4.75e-05*
	(5.06e-05)	(4.90e-05)	(5.07e-05)	(2.61e-05)	(2.63e-05)	(2.60e-05)	(5.22e-05)	(4.95e-05)	(5.13e-05)	(2.79e-05)	(2.74e-05)	(2.79e-05)
Weekly dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Monthly dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Yearly dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
(p-value)	0.6152	0.718	0.584	0.4799	0.6485	0.4254	0.0126	0.0168	0.0181	0.0052	0.004	0.0061
Observations	778	778	778	778	778	778	778	778	778	778	778	778

tandard errors of Newey and West *** p<0.01, ** p<0.05, * p<0.1.</p>

Table 2: Drivers of the day-ahead base load and peak load premium

Most of the effects are not significant in any regression. For base load, only spot price standard deviation and the wind margin shock (also suggested by AIC) are significant in several regressions (wind margin shock only when omitting variables not suggested by AIC). No significant solar shock is found, as stated above. Both significant effects are positive. The argument for a positive wind margin shock is the same as for month-ahead premiums. The positive spot price standard deviation effect as a volatility effect stands in contrast to the results of Bessembinder and Lemmon (2002), as stated in section 2, but in line with Redl and Bunn (2013) as stated in section 3.2. Due to the convexity of the merit order, shocks creating high volatility and skewness are similar and should have similar signs. However, there is no significant skewness effect on day-ahead premiums.

For peak load, only the wind margin shock and solar margin are significant (again wind margin shock only when omitting variables not suggested by AIC). The wind margin shock effect is still positive. The solar margin effect is only slightly significant in one peak load regression, and the positive sign is contrary to Redl and Bunn (2013) and counter-intuitive.

For both base and peak load, all effects are in principle zero, even if they are significant. For base load, even the null of the F-test that all explanatory variables are mutually zero cannot be rejected. Reducing the maturity of the forward contract substantially from one month to one day is likely to lead to almost perfect approximations of the expected spot price at the time when the day-ahead forward contract is determined (see also section 3.3). This should lead to forward premiums as well as effects on the forward premium very close to zero, although Wald-test results shown in section 3.3 present significant non-zero day-ahead premium means.

Table A.14 in Appendix A indicates consistent estimations of ex-ante premiums because of zero mean day-ahead regression residuals.

5. Robustness Checks

Table A.3 in Appendix A shows effects on month-ahead base and peak load premiums when the margin variables for wind and hydro power are included separately and related margin shock variables are omitted. For both base and peak load, the wind margin shock, if included, is significant in all regressions. Comparing the different base and peak load regressions separately, the effects are very similar. The wind margin variable is not significant in any regression when the wind margin shock variable is not included. This indicates that the effect of the wind margin shock variable indeed refers to unobservable wind power shocks occurring at the delivery date and not to other unspecified shocks also captured by this variable.

Regarding hydro power, in the base load regressions, the hydro margin variable is significant when the margin shock variable is omitted. It turns into an insignificant driver when the margin shock variable is included as well. Therefore, in contrast to wind power, there may be unspecified shocks in the hydro margin shock variable that do not refer to hydro shocks. Thus, interpretation of the hydro margin shock variable is not clear.

The spot price kurtosis as well as the spot price mean effect are significant and similar in all base and peak regressions. The standard deviation effect is insignificant in the regressions. The basis effect is insignificant in several regressions. This might be due to omitting wind or hydro margin shocks. Table A.4 in Appendix A shows regression results for day-ahead premiums when the wind margin variable is included separately and together with its shock. All insignificant drivers, as stated in section 4.2, are omitted. The coefficient signs have not changed. Spot price standard deviation is insignificant in most regressions. The wind margin variable turns from significance to insignificance when its shock is included as well. Thus, the wind margin shock variable might not only capture wind shocks but other unspecified shocks as well. However, day-ahead premiums near zero should lead to effects on them near zero at all as stated in section 4.2. For base load again the null of the F-test that all explanatory variables are mutually zero cannot be rejected.

Table A.5 in Appendix A shows regression results for month-ahead baseand peak load premiums when the margin and margin shock variables due to wind and hydro power are decomposed into single wind, hydro, and load and their shocks. Wind and hydro shocks are still positive significant effects on base- and peak load premiums (again the wind shock effect is larger on the peak load premium), but the load shock itself is insignificant in all regressions. The driver of the wind margin shock variable is indeed the wind shock. The hydro margin shock should be interpreted with caution as stated above. Margin variables are still insignificant as presented in section 4.1. Table A.6 in Appendix A shows related regression results for the day-ahead premiums. Again only the variables likely to be significant are included and, therefore, solar is also omitted. The wind shock effect is positive and significant, but the load shock is insignificant in both regressions. Therefore, the wind margin shock might be again driven by the wind shock. Wind itself is insignificant in both regressions, but load itself is significant for the peak premium (with a counter intuitive sign, cf. Redl and Bunn (2013)). The effects are again in principle zero, which is likely due to day-ahead forward premiums very close to zero as stated above and in section 4.2. Again, for base load, the null of the F-test that all drivers are mutually zero cannot be rejected.

Weron and Zator (2014) stated that there may be simultaneity problems if the current spot price is included on the right-hand side of the regressions because the forward price on the left-hand side might influence the current spot price. However, the forward premium as the dependent variable is defined as the difference between two different random variables (current forward and future spot price). Moreover, the future spot price is constructed as the mean of all hourly random spot prices in the future month. Finally this difference is divided by the future spot price (cf. sections 3.2 and 3.3). Therefore, this constructed random variable, the forward premium, should not be the same stochastic process as the single current forward price. Thus, simultaneity is unlikely the case. This is also clearly shown by t-test results of significantly no equal means of premiums and spot prices (cf. table A.15 in Appendix A). According again to Weron and Zator (2014), significance test results after OLS estimations may be biased due to autocorrelated conditional heteroskedastic (ARCH) residuals. Table A.16 in Appendix A shows related test results for month- and, respectively, day-ahead premiums after applying the LM-test of Engle (1982) for ARCH residuals. The null of no ARCH residuals cannot be rejected for month-ahead premiums or for the day-ahead base load premium. For the day-ahead peak load premium, there may be ARCH residuals. However, the significance level of 9.1 % for rejecting the null is high.¹⁰ Moreover, as stated above, day-ahead premiums and related effects on them are in principle zero.

6. Conclusion

The main aim in this paper has been to empirically analyze the effect of intermittent renewable supply on forward premiums in German electricity markets. Significant positive wind shock effects on both month-ahead and day-ahead ex-post forward premiums are found for German electricity markets using OLS. The effects are in line with empirical studies on other countries as well as with results from theoretical studies that consider a feedin tariff and/or high renewable penetration. Control effects reported here are also in line with past literature. The wind shock as a specific supply-type shock shifts the supply curve to the right due to the merit order effect at the time when the spot price is realized. Therefore, forward premiums should

¹⁰In Weron and Zator (2014) test results and significance levels are not presented.

rise. The wind shock effect on day-ahead forward premiums is substantially lower, i.e., close to zero. Shrinking the maturity of the forward contract substantially to the very near future leads likely to almost perfect approximations of the expected spot price at the time when the forward contract is determined. This should lead to forward premiums as well as effects on the forward premium near zero. No significant solar shock effect on daily premiums is found. The wind shock effect on month-ahead peak load premiums is larger than on base load. This should be due to higher differences in marginal costs at the right of the merit order curve and, therefore, to a larger spot price decrease.

A rise in forward premiums can be interpreted as a risk mark-up on forward prices resulting from higher intermittent wind feed-in, as stated in section 4.1. The EEX has introduced a new wind power future (see section 1).¹¹ The findings in this paper underline the need to create a new instrument dealing with risks of wind power for month-ahead forward contracts in order to reduce the stated risk mark-up for participants in forward markets as much as possible.¹² However, such an instrument might not be necessary for day-ahead contracts because in principle no effects and therefore also no wind shock effects on day-ahead premiums are found. This is at least partly in line with the EEX strategy, which is creating such an instrument for con-

¹¹https://www.eex.com/en/products/energiewende-products/ wind-power-futures

¹²Gersema and Wozabal (2016) found that market actors benefit from wind power futures (see section 2).

tracts with maturities of weeks, months, quarters, or years, but not days. There may be also solar shock effects on month-ahead premiums. However, in the present paper, that analysis has not been possible due to data availability. This might be a fruitful area for further research. Other control variables as in Redl and Bunn (2013) such as forward premium for gas are not included due to lack of data, which further research could also account for. Renewable forecast variables could be included to identify the share of forecast errors in the shock variables (see section 4.1). Dynamic behavior of wind and solar power influencing spot prices (cf. Paschen (2016)) could also be taken into account in future analyses of forward premiums. Moreover, future research could construct month-ahead premiums as the difference between the one month forward price and the corresponding realized future spot price on every day in a month leading to month-ahead premiums on a daily basis. This approach would avoid the problem that the one month forward price of the last trading day in a month is compared with the average spot price of the whole next month and not with, for example, the daily average of hourly spot prices that will be set exactly one month later. Despite the problem of interpreting ex-post premiums (see sections 2 and 3.1), the results found for month-ahead premiums in this paper seem to be reasonable. Further research could, nevertheless, focus on addressing this issue more specifically. Effects due to wind and solar shocks could, for example, be included in the model of Fleten et al. (2015) for direct analysis of forward contracts.

References

- Akaike, H., 1974. A new look at the statistical model identification. IEEE Transactions on Automatic Control 19, 716–723.
- Allaz, B., Vila, J. L., 1993. Cournot competition, forward markets and efficiency. Journal of Economic Theory 59, 1–16.
- BDEW, 2016. Erneuerbare Energien und das EEG: Zahlen, Fakten, Gafiken (2016), Bundesverband der Energie- und Wasserwirtschaft.
- Benth, F. E., Cartea, A., Kiesel, R., 2008. Pricing forward contracts in power markets by the certainty equivalence principle: Explaining the sign of the market risk premium. Journal of Banking and Finance 32, 2006–2021.
- Bessembinder, H., Lemmon, M. L., 2002. Equilibrium pricing and optimal hedging in electricity forward markets. The Journal of Finance 57 (3), 1347–1382.
- Botterud, A., Kristiansen, K., Ilic, M. D., 2010. The relationship between spot and futures prices in the Nord Pool electricity market. Energy Economics 32, 967–978.
- Cludius, J., Hermann, H., Matthes, F. C., Graichen, V., 2014. The merit order effect of wind and photovoltaic electricity generation in Germany

2008-2016: Estimation and distributional implications. Energy Economics44, 302–313.

- Dickey, D. A., Fuller, W. A., 1979. Distribution of the estimates for autoregressive time series with a unit root. Journal of the American Statistical Association 79, 355–367.
- Diko, P., Lawford, S., Limpens, V., 2006. Risk premia in electricity forward markets. Studies in Nonlinear Dynamics and Econometrics 10 (3), Article 7.
- Engle, R. F., 1982. Autoregressive conditional heteroscedasticity with estimates of the variance of United Kingdom inflation. Econometrica 50, 987–1007.
- Fischer, C., 2010. Renewable portfolio standards: When do they lower energy prices? The Energy Journal 31 (1), 101–119.
- Fleten, S. E., Hagen, L. A., Nygård, M. T., Smith-Sivertsen, R., Sollie, J. M., 2015. The overnight risk premium in electricity forward contracts. Energy Economics 49, 293–300.
- Gersema, G., Wozabal, D., 2016. An equilibrium pricing model for wind power futures, Discussion Paper, available at http://ssrn.com/abstract=2830574.
- Jensen, S. G., Skytte, K., 2002. Interactions between the power and green certificate markets. Energy Policy 30, 425–435.

- Judge, G. G., Griffiths, W. E., Hill, R. C., Lütkepohl, H., Lee, T. C., 1985. The theory and practice of econometrics. Wiley, New York.
- Kaldor, N., 1939. Speculation and economic stability. The Review of Economic Studies 7, 1–27.
- Keynes, J. M., 1930. A treatise on money. Macmillan, London.
- Kolos, S. P., Ronn, E. I., 2008. Estimating the commodity market price of risk for energy prices. Energy Economics 30 (2), 621–641.
- Newey, W. K., West, K. D., 1987. A simple, positive semi-definite, heteroskedasticity and autocorrelation consistent covariance matrix. Econometrica 55, 703–708.
- Paschen, M., 2016. Dynamic analysis of the German day-ahead electricity spot market. Energy Economics 59, 118–128.
- Peura, H., Bunn, D. W., 2016. Strategic forward trading and technology, Discussion Paper, available at http://ssrn.com/abstract=2738703.
- Redl, C., Bunn, D. W., 2013. Determinants of the premium in forward contracts. Journal of Regulatory Economics 43 (1), 90–111.
- Redl, C., Haas, R., Huber, C., Böhm, B., 2009. Price formation in electricity forward markets and the relevance of systematic forecast errors. Energy Economics 30, 356–364.

- Ritz, R. A., 2016. How does renewable competition affect forward contracting in electricity markets?, EPRG Working Paper 1610, Cambridge Working Paper in Economics 1617.
- Verbeek, M., 2008. A guide to modern econometrics. Wiley, Chichester.
- Weron, R., Zator, M., 2014. Revisiting the relationship between spot and futures prices in the Nord Pool electricity market. Energy Economics 44, 178–190.
- Woo, C. K., Moore, J., Schneiderman, B., Ho, T., Olson, A., Alagappan, L., Chawla, K., Toyama, N., Zarnikau, J., 2016. Merit-order effects of renewable energy and price divergence in California's day-ahead and realtime electricity markets. Energy Policy 92, 299–312.
- Woo, C. K., Zarnikau, J., Ho, S. T., Leung, H. Y., 2015. Virtual bidding, wind generation and California's day-ahead electricity forward premium. The Electricity Journal 28 (1), 29–48.
- Würzburg, K., Labandeira, X., Linares, P., 2013. Renewable generation and electricity prices: Taking stock and new evidence for Germany and Austria. Energy Economics 40 (Supplement 1), S159–S171.

Appendix A. Additional tables

VARIABLES	premium	premium	premium	$\operatorname{premium_peak}$	$\operatorname{premium_peak}$	premium_pea
spot	0.00699***	0.00784***	0.00678***	0.00596**	0.00699**	0.00584*
•	(0.00211)	(0.00261)	(0.00249)	(0.00294)	(0.00346)	(0.00300)
spot_std.dev.	0.00194	0.00143	0.000803	0.00466	0.00405	0.00398
-	(0.00242)	(0.00209)	(0.00294)	(0.00291)	(0.00343)	(0.00295)
spot_kurtosis	0.00120***	0.00121***	0.00138***	0.00146**	0.00147*	0.00157**
•	(0.000228)	(0.000305)	(0.000273)	(0.000630)	(0.000742)	(0.000640)
margin_wind	-0.481	-0.0167	-0.249	-0.413	0.145	-0.275
	(0.338)	(0.371)	(0.282)	(0.399)	(0.452)	(0.401)
margin_wind_shock	1.416***	. ,	1.386***	1.702***	. ,	1.685***
	(0.197)		(0.229)	(0.339)		(0.346)
margin_hydro	-1.663	-2.307	2.733*	-0.163	-0.938	2.460
	(1.380)	(1.513)	(1.541)	(2.294)	(2.699)	(1.881)
margin_hydro_shock	6.809***	6.573***		4.063^{*}	3.780	
	(1.180)	(1.247)		(2.118)	(2.496)	
basis	0.00717***	0.00524**	0.00244	0.00575*	0.00342	0.00293
	(0.00243)	(0.00235)	(0.00250)	(0.00330)	(0.00386)	(0.00302)
Constant	-0.591***	-0.526***	-0.493***	-0.328*	-0.249	-0.269
	(0.118)	(0.148)	(0.122)	(0.170)	(0.200)	(0.171)
Quarterly dummies	Yes	Yes	Yes	Yes	Yes	Yes
Yearly dummies	Yes	Yes	Yes	Yes	Yes	Yes
F(p-value)	0.000	0.000	0.000	0.000	0.000	0.000
R-squared				0.621	0.465	0.598
Observations	79	79	79	79	79	79

Standard errors of Newey and West (1987) in parentheses in base load regression in order to control for autocorrelated residuals. Non-robust standard errors in parentheses in peak load regression. *** p<0.01, ** p<0.05, * p<0.1.

Table A.3: OLS month-ahead base and peak load regressions, margin variables with and without shocks

	VARIABLES	premium	premium	premium_peak	premium_pea
		1 50 00	1.05 0.08	1.40.00	1.05 0.0
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	spot_std.dev.				1.65e-06
(5.28e-05) (5.30e-05) (3.82e-05) (7.34e-05) (9.734e-05) (4.60e-05) (4.60e-05) (1.60e-05) (1.60e-05) (1.60e-05) (1.60e-05) Constant -3.66e-05 -3.97e-05 -3.89e-05 - (2.77e-05) (2.77e-05) (1.60e-05) (2.77e-05) (1.60e-05) (2.77e-05)					(1.30e-06)
margin_wind_shock 7.34-05 7.34-05 9.05 Constant (4.60-05) -3.97-05 -3.89e-05 -3.97-05 (2.74e-05) (2.69-05) (2.77e-05) (2	margin_wind	8.78e-05*	4.27e-05	6.49e-05*	4.09e-06
(4 60e-05) (Constant -3.66e-05 -3.97e-05 -3.89e-05 -2.77e-05 (Weddy dumnies Yes Yes Yes Yes Yes Monthly dummies Yes Yes Yes Yes Yes Yearly dummies Yes Yes Yes Yes Yes (Parly dummies) 0.011 0.6293 0.0087 10007		(5.28e-05)	(5.30e-05)	(3.82e-05)	(4.68e-05)
Constant -3.66e-05 -3.97e-05 -3.89e-05 - (2.74e-05) (2.69e-05) (2.77e-05) (2.77e-05) (2.77e-05) (2.77e-05) Woekly dummies Yes Yes Yes Yes Yes Monthly dummies Yes Yes Yes Yes Yes Yeardy dummies Yes Yes Yes Yes Yes Flow-able 0.0611 0.6290 0.0087 1 0.0087	margin_wind_shock		7.34e-05		9.88e-05*
(2.74e-05) (2.69e-05) (2.77e-05) (Weekly dummies Yes Yes Yes Monthly dummies Yes Yes Yes Yearly dummies Yes Yes Yes P(p-value) 0.0111 0.6293 0.0087			(4.60e-05)		(5.21e-05)
Weekly dummies Yes Yes Yes Yes Monthly dummies Yes Yes Yes Yes Yearly dummies Yes Yes Yes Yes F(p-value) 0.6111 0.6293 0.0067	Constant	-3.66e-05	-3.97e-05	-3.89e-05	-4.31e-05
Monthly dummies Yes Yes Yoarly dummies Yes Yes (Py-value) 0.0111 0.6293 0.0087		(2.74e-05)	(2.69e-05)	(2.77e-05)	(2.77e-05)
Yearly dummies Yes Yes Yes F(p-value) 0.6111 0.6293 0.0087	Weekly dummies	Yes	Yes	Yes	Yes
F(p-value) 0.6111 0.6293 0.0087	Monthly dummies	Yes	Yes	Yes	Yes
	Yearly dummies	Yes	Yes	Yes	Yes
	F(p-value)	0.6111	0.6293	0.0087	0.0044
Ubservations 778 778 778	Observations	778	778	778	778

Table A.4: OLS day-ahead base and peak load regressions, wind margin variable with and without shocks

VARIABLES	premium	premium_peak	
spot	0.00586**	0.00420	
•	(0.00222)	(0.00295)	
spot_std.dev.	0.00222	0.00478	
•	(0.00223)	(0.00301)	
spot_kurtosis	0.00127***	0.00159**	
•	(0.000235)	(0.000630)	
wind	-0.00846	-0.00711	
	(0.00553)	(0.00631)	
wind_shock	0.0232***	0.0277***	
	(0.00293)	(0.00541)	
hydro	-0.0288	-0.00218	
5	(0.0214)	(0.0388)	
hydro_shock	0.123***	0.0827**	
5	(0.0249)	(0.0358)	
load	-0.00188	-0.00501	
	(0.00313)	(0.00486)	
load_shock	-0.00278	0.00134	
	(0.00228)	(0.00360)	
basis	0.00518**	0.00228	
	(0.00248)	(0.00373)	
Constant	-0.281	-0.0675	
	(0.247)	(0.321)	
Quarterly dummie	es Yes	Yes	
Yearly dummies	Yes	Yes	
F(p-value)	0.000	0.000	
R-squared		0.652	
Observations	79	79	

Standard errors of Newey and West (1987) in parentheses in base load regression in order to control for autocorrelated residuals. Non-robust standard errors in parentheses in peak load regression. *** p<0.01, ** p<0.05, * p<0.1.

Table A.5: Single wind, hydro and load shocks on month-ahead base and peak load premium

VARIABLES	premium	premium_peak	
spot_std.dev.	2.37e-06**	1.89e-06	
opor_oralde11	(1.15e-06)	(1.27e-06)	
wind	3.07e-07	-1.39e-07	
	(8.39e-07)	(7.56e-07)	
wind_shock	1.31e-06*	$1.61e-06^*$	
	(7.66e-07)	(8.41e-07)	
load	-7.78e-07	-3.41e-06*	
	(2.32e-06)	(1.83e-06)	
load_shock	-2.72e-06	1.28e-06	
	(2.37e-06)	(2.18e-06)	
Constant	0.000171	7.87e-05	
	(0.000120)	(7.73e-05)	
weekly dummies	Yes	Yes	
monthly dummie	es Yes	Yes	
Yearly dummies	Yes	Yes	
F(p-value)	0.611	0.0017	
Observations	778	778	

Standard errors of Newey and West (1987) in parentheses in order to control for autocorrelated and heteroskedastic residuals. *** p<0.01, ** p<0.05, * p<0.1.

Table A.6: Single wind and load shocks on day-ahead base and peak load premium

Variable	Obs	Mean	Std. Dev.	Min	Max	Test (p-value)
premium	79	.021	.086	159	.313	.0368
premium_peak	79	.24	.105	.035	.687	.000
spot (€/MWh)	79	39.873	7.649	25.358	56.831	-
spot_std.dev. (€/MWh)	79	14.139	4.558	6.311	38.719	-
spot_skewness	79	537	1.364	-8.422	2.057	-
spot_kurtosis	79	7.94	18.097	2.07	156.359	-
margin_wind	79	.092	.038	.037	.233	-
margin_hydro	79	.043	.007	.03	.066	-
basis (€/MWh)	79	.689	3.635	-9.803	11.292	-
Wald test (cf. Judge et a	l. (198	5)) in las	t column wi	th null h	ypothesis:	Variable has zero mean.

Wind and hydro margin shocks are omitted because the series are one month future series of the margin variables, cf. equation 3.

Table A.7: Descriptive statistics of monthly variables

Variable	Obs	Mean	Std. Dev.	Min	Max	Test (p-value)
premium	778	0	0	0	0	0
premium_peak	778	0	0	0	0	0
spot	778	38.383	8.521	4.453	72.069	
spot_std.dev.	778	10.506	3.781	3.094	29.102	
spot_skewness	778	.012	.611	-1.837	2.861	-
spot_kurtosis	778	2.295	.982	1.192	12.233	
margin_wind	778	.107	.09	.006	.476	
margin_solar	778	.063	.044	.001	.183	
basis	778	-1.621	6.556	-29.669	23.313	-

Wald test (cf. Judge et al. (1985)) in last column with null hypothesis: Variable has zero mean. Wind and solar margin shocks are omitted because the series are one day future series of the margin variables, cf. equation 5.

Table A.8: Descriptive statistics of daily variables

Variables	no constant, no trend	constant, no trend	constant, trend
premium	< .01	2.620e-14	1.460e-13
premium_peak	$.01$	8.410e-12	1.137e-10
spot	> .1	.00267823	.00567495
spot_std.dev.	$.01$	5.796e-09	1.682e-07
spot_skewness	< .01	1.011e-12	2.107e-12
spot_kurtosis	< .01	2.690e-12	9.000e-12
margin_wind	> .1	.00245314	.00451365
margin_hydro	> .1	.00019365	.00479535
wind	> .1	.00168411	.00954781
hydro	> .1	.00004289	.00147819
load	> .1	.00010081	.00584689
basis	< .01	2.331e-15	1.672e-15

Test equations include trend, constant or none.

Table A.9: Augmented Dickey-Fuller test for unit roots for monthly time series

Variables	no constant, no trend	constant, no trend	constant, tren
premium	< .01	0	0
premium_peak	< .01	0	0
spot	$.01$	0	4.411e-20
spot_std.dev.	< .01	0	0
spot_skewness	< .01	0	0
spot_kurtosis	< .01	0	1.188e-22
margin_wind	< .01	0	6.745e-22
margin_solar	< .01	9.204e-14	1.041e-09
wind	< .01	0	1.123e-21
solar	< .01	4.630e-14	5.442e-10
load	> .1	9.202e-11	3.028e-08
basis	< .01	0	0

Test equations include trend, constant or none.

rest equations include trend, constant of none.

Table A.10: Augmented Dickey-Fuller test for unit roots for daily time series

Variables	spot	spot_std.dev.	spot_skewness	spot_kurtosis	margin_wind	margin_wind_shock	margin_hydro	margin_hydro_shock	wind	wind_shock	hydro	hydro_shock	load	load_shock	basis
spot	1.000														
spot_std.dev.	0.100	1.000													
spot_skewness	0.109	-0.462	1.000												
spot_kurtosis	-0.025	0.523	-0.832	1.000											
margin_wind	-0.415	0.184	0.029	-0.029	1.000										
margin_wind_shock	-0.187	0.112	0.146	0.002	0.643	1.000									
margin_hydro	-0.619	-0.155	-0.014	-0.104	-0.020	-0.090	1.000								
margin_hydro_shock	-0.510	-0.218	0.157	-0.148	0.058	-0.010	0.723	1.000							
wind	-0.338	0.236	0.035	-0.026	0.989	0.637	-0.105	-0.015	1.000						
wind_shock	-0.117	0.173	0.127	0.015	0.643	0.989	-0.163	-0.093	0.651	1.000					
hydro	-0.528	0.016	0.013	-0.124	0.060	-0.026	0.913	0.665	0.028	-0.068	1.000				
hydro_shock	-0.393	-0.048	0.158	-0.145	0.151	0.068	0.596	0.913	0.115	0.038	0.643	1.000			
load	0.418	0.365	0.048	0.010	0.150	0.138	-0.563	-0.400	0.291	0.231	-0.186	-0.142	1.000		
load_shock	0.411	0.428	-0.108	0.091	0.150	0.144	-0.522	-0.551	0.254	0.283	-0.285	-0.171	0.688	1.000	
basis	-0.111	0.150	-0.444	0.158	0.040	-0.193	0.139	-0.255	0.024	-0.154	0.096	-0.200	-0.161	0.220	1.000

Table A.11: Cross-correlations of explanatory monthly variables

Variables	spot	spot_std.dev.	spot_skewness	spot_kurtosis	margin_wind	margin_wind_shock	margin_solar	margin_solar_shock	wind	wind_shock	solar	solar_shock	load	load_shock	basis
spot	1.000														
spot_std.dev.	0.484	1.000													
spot_skewness	0.087	0.031	1.000												
spot_kurtosis	-0.083	0.215	0.575	1.000											
margin_wind	-0.624	0.054	-0.121	0.101	1.000										
margin_wind_shock	-0.419	-0.018	-0.038	0.024	0.690	1.000									
margin_solar	-0.309	-0.520	0.086	0.008	-0.272	-0.231	1.000								
margin_solar_shock	-0.293	-0.506	0.036	-0.006	-0.255	-0.257	0.889	1.000							
wind	-0.604	0.051	-0.103	0.098	0.993	0.693	-0.300	-0.283	1.000						
wind_shock	-0.402	0.003	-0.031	0.029	0.694	0.987	-0.265	-0.295	0.709	1.000					
solar	-0.316	-0.526	0.117	0.026	-0.266	-0.226	0.994	0.882	-0.290	-0.256	1.000				
solar_shock	-0.302	-0.501	0.054	0.016	-0.247	-0.260	0.885	0.986	-0.270	-0.285	0.887	1.000			
load	0.150	0.156	0.189	0.035	0.116	0.129	-0.527	-0.514	0.207	0.221	-0.458	-0.451	1.000		
load_shock	0.131	0.234	0.096	0.064	0.082	0.032	-0.391	-0.457	0.147	0.156	-0.345	-0.338	0.737	1.000	
basis	-0.298	0.055	-0.091	0.094	0.284	-0.272	0.006	-0.078	0.280	-0.226	0.009	-0.017	0.022	0.378	1.000

Table A.12: Cross-correlations of explanatory daily variables

	mean	test statistic	p-value
base load	-8.598e-11	1.972e-16	.999999999
peak load	-9.431e-10	1.676e-16	.999999999
Null hypothesis: Residuals have zero mean.			

Table A.13: Wald test (cf. Judge et al. (1985)) for zero mean of month-ahead regression residuals (base and peak load (see section 4.1))

	mean	test statistic	p-value
base load	-6.73e-14	0	1
peak load	1.25e-14	0	1
Null hypothesis: Residuals have zero mean.			

Table A.14: Wald test (cf. Judge et al. (1985)) for zero mean of day-ahead regression residuals (base and peak load (see section 4.2))

month-ahead	base load	peak load
		1
Test statistic	-45.996909	-45.741273
p-value	8.338e-59	1.260e-58
day-ahead	base load	peak load
Test statistic	-125.64291	-125.64292
p-value	0	0
Two sided t-test for prem	nium mean and spot price mean equality.	
Null hypothesis: Means a	re equal.	
Test statistic is student's	t.	
Equal variances are not a	ssumed.	

Table A.15: Two sided t-test for premium mean and spot price mean equality

month-ahead	base load	peak load
Test statistic	.022	.045
p-value	.8831	.8314
day-ahead	base load	peak load
Test statistic	2.33	2.855
p-value	.1269	.0911
Engle LM-test	for ARCH 1	residuals after month- and day-ahead base and peak load regressions (see section 4).
Null hypothes	is: No ARCI	I residuals.

Test statistic asymptotically chi-squared.

Table A.16: Engle LM-test for ARCH residuals

Zuletzt erschienen /previous publications:

V-397-16	Marius Paschen, The effect of intermittent renewable supply on the forward
V-396-16	premium in German electricity markets Heinz Welsch, Philipp Biermann, Poverty is a Public Bad: Panel Evidence from Subjective Well being Date
V-395-16	Subjective Well-being Data Philipp Biermann , How Fuel Poverty Affects Subjective Well-Being: Panel Evidence from Germany
V-394-16	Heinz Welsch , Electricity Externalities, Siting, and the Energy Mix: A Survey
V-393-16	Leonhard Kähler, Klaus Eisenack, Strategic Complements in International
V-575-10	Environmental Agreements: a New Island of Stability
V-392-16	Christoph Böhringer, Xaquin Garcia-Muros, Ignacio Cazcarro, Iñaki Arto, The Efficiency Cost of Protective Measures in Climate Policy
V-391-16	Achim Hagen, Juan-Carlos Altamirano-Cabrera, Hans-Peter Weikard, The
	Influence of Political Pressure Groups on the Stability of International Environmental
	Agreements
V-390-16	Christoph Böhringer, Florian Landis, Miguel Angel Tovar Reaños, Cost-
	effectiveness and Incidence of Renewable Energy Promotion in Germany
V-389-16	Carsten Helm, Mathias Mier, Efficient diffusion of renewable energies: A roller-
	coaster ride
V-388-16	Christoph Böhringer, Jan Schneider, Emmanuel Asane-Otoo, Trade In Carbon and The Effectiveness of Carbon Tariffs
V-387-16	Achim Hagen, Leonhard Kähler, Klaus Eisenack, Transnational Environmental
	Agreements with Heterogeneous Actors
V-386-15	Jürgen Bitzer, Erkan Gören, Sanne Hiller, Absorption of Foreign Knowledge:
11.005.15	Firms' Benefits of Employing Immigrants
V-385-15	Klaus Eisenack, Julien Minnemann, Paul Neetzow, Felix Reutter, Contributions
V 204 15	to the institutional economics of the energy transition
V-384-15	Christoph Böhringer, Xaquín Garcia-Muros, Mikel Gonzalez-Eguino, Luis Rey,
V 202 15	US Climate Policy: A Critical Assessment of Intensity Standards
V-383-15	Christoph Böhringer, Edward J. Balistreri, Thomas F. Rutherford, Carbon
V-382-15	policy and the structure of global trade Christoph Böhringer, Brita Bye, Taran Fæhn, Knut Einar Rosendahl, Output-
v-362-13	based rebating of carbon taxes in the neighbor's backyard
V-381-15	Christoph Böhringer, Markus Bortolamedi , Sense and No(n)-Sense of Energy
v-301-13	Security Indicators
V-380-15	Christoph Böhringer, Knut Einar Rosendahl, Halvor Briseid Storrøsten,
1 000 10	Mitigating carbon leakage:Combining output-based rebating with a consumption tax
V-379-15	Jan Micha Steinhäuser, Klaus Eisenack, Spatial incidence of large-scale power
	plant curtailment costs
V-378-15	Carsten Helm, Franz Wirl, Climate policies with private information: The case for
	unilateral action
V-377-15	Klaus Eisenack, Institutional adaptation to cooling water scarcity in the electricity
	sector under global warming
V-376-15	Christoph Böhringer, Brita Bye, Taran Fæhn, and Knut Einar Rosendahl,
	Targeted carbon tariffs – Carbon leakage and welfare effects
V-375-15	Heinz Welsch, Philipp Biermann, Measuring Nuclear Power Plant Externalities
	Using Life Satisfaction Data: A Spatial Analysis for Switzerland
V-374-15	Erkan Gören, The Relationship Between Novelty-Seeking Traits And Comparative
	Economic Development
V-373-14	Charlotte von Möllendorff, Heinz Welsch, Measuring Renewable Energy
	Externalities: Evidence from Subjective Well-Being Data
V-372-14	Heinz Welsch, Jan Kühling, Affective States and the Notion of Happiness: A
	Preliminary Analysis
V-371-14	Carsten Helm, Robert C. Schmidt, Climate cooperation with technology
	investments and border carbon adjustment

V-370-14	Christoph Böhringer, Nicholas Rivers, Hidemichi Yonezawa, Vertical fiscal
	externalities and the environment
V-369-14	Heinz Welsch, Philipp Biermann, Energy Prices, Energy Poverty, and Well-Being:
	Evidence for European Countries
V-368-14	Marius Paschen, Dynamic Analysis of the German Day-Ahead Electricity Spot
	Market
V-367-14	Heinz Welsch, Susana Ferreira, Environment, Well-Being, and Experienced
	Preference
V-366-14	Erkan Gören, The Biogeographic Origins of Novelty-Seeking Traits
V-365-14	Anna Pechan, Which Incentives Does Regulation Give to Adapt Network Infrastructure to
	Climate Change? - A German Case Study
V-364-14	Christoph Böhringer, André Müller, Jan Schneider, Carbon Tariffs Revisited
V-363-14	Christoph Böhringer, Alexander Cuntz, Diemtar Harhoff, Emmanuel A. Otoo,
	The Impacts of Feed-in Tariffs on Innovation: Empirical Evidence from Germany
V-362-14	Christoph Böhringer, Nicholas Rivers, Thomas Ruhterford, Randall Wigle,
	Sharing the burden for climate change mitigation in the Canadian federation
V-361-14	Christoph Böhringer, André Müller, Environmental Tax Reforms in Switzerland A
	Computable General Equilibrium Impact Analysis
V-360-14	Christoph Böhringer, Jared C. Carbone, Thomas F. Rutherford,
	The Strategic Value of Carbon Tariffs
V-359-13	Heinz Welsch, Philipp Biermann, Electricity Supply Preferences in Europe:
	Evidence from Subjective Well-Being Data
V-358-13	Heinz Welsch, Katrin Rehdanz, Daiju Narita, Toshihiro Okubo, Well-being
	effects of a major negative externality: The case of Fukushima
V-357-13	Anna Pechan, Klaus Eisenack, The impact of heat waves on electricity spot markets
V-356-13	Heinz Welsch, Jan Kühling, Income Comparison, Income Formation, and
	Subjective Well-Being: New Evidence on Envy versus Signaling
V-355-13	Christoph Böhringer, Knut Einar Rosendahl, Jan Schneider, Unilateral Climate
	Policy: Can Opec Resolve the Leakage Problem?
V-354-13	Christoph Böhringer, Thomas F. Rutherford, Marco Springmann; Clean-
	Development Investments: An Incentive-Compatible CGE Modelling Framework
V-353-13	Erkan Gören, How Ethnic Diversity affects Economic Development?
V-352-13	Erkan Gören, Economic Effects of Domestic and Neighbouring Countries'Cultural
11 051 10	Diversity
V-351-13	Jürgen Bitzer, Erkan Gören, Measuring Capital Services by Energy Use: An
V 250 10	Empirical Comparative Study
V-350-12	Heinz Welsch, Jan Kühling, Competitive Altruism and Endogenous Reference
V 240 12	Group Selection in Private Provision of Environmental Public Goods
V-349-12	Heinz Welsch, Organic Food and Human Health: Instrumental Variables Evidence
V-348-12	Carsten Helm, Dominique Demougin, Incentive Contracts and Efficient
V-347-12	Unemployment Benefits in a Globalized World
V-347-12	Christoph Böhringer, Andreas Lange, Thomas F. Rutherford, Optimal Emission
	Pricing in the Presence of International Spillovers: Decomposing Leakage and Terms-
V-346-12	of-Trade Motives Christenh Böhringer, Jared C. Cerbang, Themas F. Butherford, Efficiency and
v-340-12	Christoph Böhringer, Jared C. Carbone, Thomas F. Rutherford , Efficiency and Equity Implications of Alternative Instruments to Reduce Carbon Leakage
V-345-12	Christoph Böhringer, Brita Bye, Taran Fæhn, Knut Einar Rosendahl,
v-343-12	Alternative Designs for Tariffs on Embodied Carbon: A Global Cost-Effectiveness
	Analysis
V-344-12	Klaus Eisenack und Leonhard Kähler, Unilateral emission reductions can lead to
v-344-12	Pareto improvements when adaptation to damages is possible
V-343-11	Heinz Welsch and Jan Kühling , Anti-Inflation Policy Benefits the Poor: Evidence
v-J4J-11	from Subjective Well-Being Data
V-342-11	Heinz Welsch and Jan Kühling, Comparative Economic Performance and
v-J+2-11	Institutional Change in OECD Countries: Evidence from Subjective Well-Being Data
V-341-11	Carsten Helm and Stefan Pichler , Climate Policy with Technology Transfers and
v-J=1=11	Permit Trading